Author's response to reviews

Title: Stimulation of the autonomic nervous system in colorectal surgery: a study protocol for a randomized controlled trial

Authors:

Tim M.P. Berghmans (timberghmans@gmail.com)
Karel W.E. Hulsewé (k.hulsewe@orbisconcern.nl)
Wim A. Buurman (w.buurman@maastrichtuniversity.nl)
Misha D.P. Luyer (misha.luyer@catharina-ziekenhuis.nl)

Version: 3 Date: 24 May 2012

Author's response to reviews: see over
Dear Editor,

Thank you for accepting the revised version of our manuscript entitled: “Stimulation of the autonomic nervous system in colorectal surgery: study protocol for a randomized controlled trial”.

The suggested formatting corrections have been made and the study protocol is now formatted according to the Trials instructions.

Yours sincerely,

Misha Luyer